Author’s response to reviews

Title: Factors influencing disability in patients with chronic low back pain attending a tertiary hospital in sub-Saharan Africa

Authors:

Marie Doualla Bija (marie.doualla@gmail.com)
Jeannine Aminde (jeannineatemanyingu@rocketmail.com)
Leopold Aminde (l.aminde@uq.edu.au)
Fernando Kemta-Lekpa (fklekpa@yahoo.fr)
Felix Mangan Kwedi (kwedi80@yahoo.fr)
Emmanuel Yenshu (socpolub@hotmail.com)
Alain Chichom (alainchichom@yahoo.com)

Version: 2 Date: 02 Oct 2018

Author’s response to reviews:

Dear Editor,

We proceeded to the corrections of the abstract as requested by the reviewer.

In Line 9: « to investigate predictors » is replaced by « investigate factors associated »

In Line 19: « BBDS predicted greater disability » is replaced by « BBDS was associated to greater disability »